Author's response to reviews

Title: Assessment of the paraspinal muscles of subjects presenting an idiopathic scoliosis: an EMG pilot study.

Authors:

Nathaly Gaudreault (nathalygaudreault@yahoo.ca)
Bertrand Arsenault (bertrand.arsenault@umontreal.ca)
Christian Lariviere (lariviere.christian@irsst.qc.ca)
Sophie J De Serres (sophie.deserres@mcgill.ca)
Charles-Hilaire Rivard (crivard@justine.umontreal.ca)

Version: 5 Date: 6 March 2005

Author's response to reviews:

Dear Editor
Here is the final version of the manuscript with the formatting changes done according to your request.
Thank you for considering our manuscript for publication
Nathaly Gaudreault, M.Sc., PT